Author’s response to reviews

Title: Planning dietary improvements without additional costs for low-income individuals in Brazil: linear programming optimization as a tool for public policy in nutrition and health

Authors:

Eliseu Verly-Jr (eliseujunior@gmail.com)
Rosely Sichieri (rosely.sichieri@gmail.com)
Nicole Darmon (nicole.darmon@inra.fr)
Matthieu Maillot (matthieu.maillot@ms-nutrition.com)
Flavia Sarti (flamori@usp.br)

Version: 3 Date: 08 Jul 2019

Author’s response to reviews:

Dear Editor,

Thanks for the opportunity to revise the manuscript. We double checked the writing and made some correction including those pointed out in your comments. The modifications are highlighted in the text. Particularly: 102 requires clarification - what happened to states that were not categorized?

We made it clearer that all states were included, but not all of them were stratified into four but three geographic areas. Please see the lines 102-103.